

An Efficient Cluster Head Selection for Wireless Sensor Network-Based Smart Agriculture Systems

Mehdi Gheisari

EasyChair preprints are intended for rapid dissemination of research results and are integrated with the rest of EasyChair.

June 10, 2022

An Efficient Cluster Head Selection for Wireless Sensor Network-based Smart Agriculture Systems

Mehdi Gheisari¹

Abstract— With the increasing availability of high-resolution satellite and drone images and the Internet of Things (IoT) has begun transforming remote sensing of agriculture by improving accessibility and frequency of updates. Modern IoT-based smart agriculture systems use Wireless Sensor Networks (WSNs) to gather information from an ecosystem that regulates the quantity of water in agricultural fields could be one of these activities. The WSNs remained a challenge to transfer data to drones for analysis purposes. These are composed of tiny sensory architectures organized together to bring efficiency and scalability features to a network. WSN nodes are controlled and managed by a cluster. It is quite difficult to design an efficient leader election protocol. The computation power, storage space, and energy supply of sensor nodes make them unable to frequently switch to a different cluster. The WSN cluster-head election process requires a lot of energy (evaluation and computational process to select the most appropriate node with the least impact on network fragmentation in energy consumption of selected node). Then it is necessary to formulate a mechanism where WSNs utilize the least energy to coordinate with the remote sensing sources. This study presents a cluster election algorithm using the fuzzy logic inference system. It uses a coordinates system to map network nodes and map them based on prioritized scheduling. Lifetime augmentation in wireless sensor networks has always been of great interest. During data transmission from normal sensor nodes to the base station (sink), excess energy is dissipated. Optimizing the energy dissipation of WSNs through the selection of cluster heads is a powerful way to increase the lifespan. By electing more efficient nodes as cluster heads, the proposed method extends the network's lifetime by reducing the number of unimportant communications between nodes. With the utilization of network resources efficiently, the network's lifetime is extended. The proposed algorithm is evaluated with the LEACH (Low Energy Adaptive Clustering Structure) algorithm and FCA method based on the remaining energy and the number of active nodes. The simulation results show that the proposed algorithm utilizes less energy for communication with remote sensory equipment for intelligent agriculture. The performance of the method improved for remaining energy by 9%, the number of active nodes rate by 24%, and indirectly network resource utilization than other states of the art solutions.

Index Terms— Smart Agriculture, Fuzzy Logic, Wireless Sensors, WSN, Clustering, Decision System

1 INTRODUCTION

Remote sensing is extensively used in smart agriculture systems to manage and monitor agriculture yields both in productivity and sustainability [1,2]. These systems consist of modern communication equipment aimed at helping farmers improve crop productivity. A smart agriculture system consists of precision equipment such as sensors, actuators, and unmanned aerial vehicles. Sensors are installed on agriculture farms to monitor variations and trends. Most of the time, these sensors communicate by using wireless communication. Wireless Sensor Networks (WSN) constitute an important component of a smart agriculture system. It is extensively used both for monitoring and tracking applications [3]. It has been used widely in environments such as temperature, pressure measurement, and environment monitoring [4-5]. However, if we want to use WSN, we have to keep in mind that WSN does not have an external power supply, and sensors rely on their internal power and are almost difficult to recharge and replace. However, work has been done to compensate for this problem. Energy consumption can be reduced or compensated as much possible as [43].



Fig. 1. Remote monitoring of agriculture farms

Therefore, power consumption has become one of the most critical issues in WSN. Most researchers in this field have focused on reducing energy consumption by presenting different approaches that have greatly improved our focus on the subject. With the help of proper messaging and optimal use of network resources, we aim to reduce energy consumption. WSN architecture consists of interconnected sensor nodes [6-8]. Each sensor node is a single, tiny device equipped with a simple microcontroller. WSN-based systems are used in many fields such as industry, civilian application, healthcare, and agriculture. The biggest challenge in maintaining sensor efficiency is the complex nature of the optimized algorithms. Their complexity often constrains the WSN nodes, such as processing power and size. WSN node management is a complicated task [9-11]. Many works of literature are available on the topic of network node management [12-13]. Apart from those mentioned above, most sensor nodes detect functions such as motion, temperature, humidity, and light. These sensors use a Pico-power microcontroller. It helps control the power transceiver and the communication function of the sensors. The recovery of the captured information from dead nodes is a complicated task. Therefore, various localization methods ensure that the information arrives at the right time from the active nodes.

In remote sensing environments, agriculture farms are monitored using aerial vehicles such, e.g. drones, as shown in Fig 1. These drones monitor the farms by communicating with the cluster leader of the WSN for exchange and update of information. The cluster heads do not send information directly to aerial vehicles but rather through a base station (Sink). WSN classification is based on performance, network architecture, and user requirements. Includes creating a cluster with the features required for the classification phase [46]. Romika et al. In 2020 [22] proposed a new distributed algorithm for selecting cluster heads. This algorithm has three main features and is based on symmetric key operations.

The cluster nodes unanimously elect the leader. These are resilient against hacking attacks and have no impact on cluster leader election decisions. The node is also considered faulttolerant to ensure safety for messages lost or altered through malicious attacks. Tao et al., in 2019 [23], also presented a cluster head selection algorithm. The proposed algorithm uses fuzzy rules. It considers the section's three main characteristics: the energy available from the nodes, the number of neighboring nodes, and the distance from the base station (or sink). This method is compared to hierarchical distributed clustering. The results show that this method improves the performance in terms of network lifetime. The research proposed by Baby and Vasudevan [24] introduced a distributed and stable algorithm for cluster head election. The proposed algorithm works with the calculation of probability values. It considers two parameters, i.e., residual energy of the sensor node, and compares it with the average energy of the combined WSN node. The results show that the proposed algorithm is stable and efficient compared to traditional technology. The main disadvantage of this algorithm is that it sends or receives a relatively high number of data packets to the sink compared to other algorithms. Mirel et al. (2020) [25] Proposed a multilayer clustering algorithm. Consider three centralized algorithms that share distributed functions to reduce the number of CH (Cluster Head) candidates. The multilayer algorithm also reduces the message exchange load using efficient topology architecture.

The lifetime and stability of a WSN are ensured through a proper energy utilization method. This issue is enabled through the concept of clustering [14,15]. Clustering improves energy efficiency through cluster-head selection methods. Several ways to select cluster heads in sensor networks [16,17]. If a suitable cluster is selected, it will help reduce the transmission distance of the communication distance of the main node that is

not the cluster head. The Low Energy Adaptive Clustering Hierarchy (LEACH) protocol [18,19] is considered an important clustering method because it improves energy efficiency in a very efficient way. Besides, other swarm intelligence approaches, such as [20,21], have also improved computing power usage. This research proposes a WSN cluster leader selection algorithm that uses fuzzy logic, reducing communication overhead, improving energy efficiency, and reducing memory requirements. Fuzzy logic is one of the most effective methods for combining and evaluating parameters in wireless sensor networks since it allows us to combine and evaluate data more effectively. However, it is possible to use it to improve the performance of sensor networks, which has led us to use this logic. The author compares the proposed algorithm with the advanced algorithm (such as LEACH) in energy consumption.

The research offers a fuzzy-based system that will lead to smart agriculture and effective crop development. The model's goal is to monitor agricultural fields for effective water consumption. The suggested approach demonstrates high water utilization in irrigation and is advantageous for the restricted water area. The contributions of this study are highlighted as follows:

- a. Improve decision making by the fuzzy choice of cluster leader, reduce resource consumption, and generally improve performance;
- b. The proposed model considers both cluster leader selection and the power consumption;
- c. Evaluation and extension lifetime of the network;
- d. The network consists of homogeneous nodes with the same energy level;

1.1 WSN in agriculture

WSN is an important enabler for precision agriculture. Because it helps gather, monitor, and analyze data from agricultural areas [48]. WSN might connect IoT sensors, allowing for real-time detection of soil and climatic conditions as well as irrigation automation [49]. IoT, clustering, RFID, and unmanned aerial vehicles (UAV) are important enablers for creating smarter, more sustainable agriculture that benefits farmers and consumers. WSN solutions drew a lot of attention and gained much traction because of their potential capacity to improve farmers' awareness of crop conditions, resource utilization, and environmental conditions. The proposed method recommends improvement approaches for existing WSN deficiencies in the context of energy efficiency in order to provide an information platform for WSN to play a more important role in crop production.

The authors propose a novel cluster-head selection mechanism using fuzzy logic to establish a fully connected network with efficiency and network resource utilization in an agriculture environment. Using the proposed algorithm, full connectivity between sensor nodes is assumed to be energyefficient while enhancing network lifetime. This increases the network's longevity by reducing the dead node count considerably. The fundamental point of the proposed strategy appropriation in horticulture is information collection, environmental monitoring, and data analysis.

The rest of the study is organized as follows. Section 2 summarizes the related work. Section 3 briefly explains the methodology used in the proposed study. Section 4 reports on the evaluation, results, and discussion of the proposed system. Finally, Section 5 summarizes the article.

2 RELATED WORK

Remote sensing is the concept of using sensors to accumulate data. The sensor detects statistics by using gathering energy reflected from the ground. Sensors are often established on satellites or statistics series drones [35, 36]. Remote sensing explores treasured records due to the fact it affords an overview. Remote sensing science is active and passive, based totally on sensory responses to goal stimuli. This statistics can be used for correct estimations because it uses a large vicinity with repetitive calibrated sensory measurements to estimate changes in one-of-a-kind resolutions [37]. This issue helps accomplish correct effects in contrast to the usual facts series methods.

The world summit on food security concluded on a note that the world population would grow by 10 billion by the year 2050 [38]. With the evaluated rate of growth, agriculture production must also increase. This issue will help in accommodating the increasing food demands. Agricultural land management can improve agriculture production by improving water resources, land up-grading, and crop monitoring. In modern agriculture systems, monitoring is performed through remote sensing techniques. Remote sensing in agriculture is of fundamental importance because it provides ways and means to monitor agriculture yields and crops effectively. Agriculture systems involving remote sensing technologies are widely known as smart agriculture systems. These systems involve sensors and sensory equipment for monitoring agriculture crops. These sensors are connected to exchange information using a cluster head sensor [39, 40]. A cluster head later communicates information to a base station, which can be based or mounted on a drone or satellite.

This section discusses the latest developments in WSN cluster leader selection. In [29], the LEACH algorithm was proposed. It was developed to achieve energy efficiency in a clustered hierarchical network. This methodology proposes to use a dedicated central location for storing and collecting cluster data. Each cluster has a specified location where data collection is performed. It also uses probabilistic methods to select cluster heads; however, it does not consider the remaining energy sources of the cluster. This issue is because information can only be obtained after transmitting and receiving information. However, this is a complex and energyconsuming task. In [30], an improved LEACH algorithm for cluster head selection was proposed. To solve the problem of selecting the head of the cluster uses the time division multiple access (TDMA) method. The base station updates the nominated cluster head to synchronize future transmission. The cluster head also checks its immediate neighbors to reduce future data and time-related delays. The presented LEACH-C performs the same cluster head selections for other components of the networks to find sensor nodes with a smaller number of computational power and communication resources. Performing all computing tasks at the base station would result in higher energy consumption, adversely affecting its performance.

Several research proposals target improving the performance of cluster head selection [1-3, 23, 27]. Cluster head selection is given more importance because it helps in improving the computational complexity of the wireless sensor network. Several particle swarm optimizations (PSO) based approaches are presented to address these challenges. PSO concepts are vastly used in computing to optimize challenges by relating their availability to given quality value.

Swarm intelligence has often been used in WSN protocols for optimizing energy-related constraints. PSO-C was implemented by Abdul Latiff et al. 2007 [31] in WSNs to determine optimal cluster head sections for clusters. By using PSO-C, location cluster heads can achieve effective results, thus conserving energy and improving the lifetime of sensors. Unfortunately, the selection process does not consider the cluster head distance concerning the base station. Ignoring the distance parameter may affect the energy efficiency and data transmission time. Srinivasa et al (2017) [32], addressed this challenge. The author proposed a PSO-based Power Saving Cluster Head Selection (PSOECHS) algorithm. It uses an objective function that measures the cluster head and base station distance. Then it uses an objective function to select a node within each cluster.

PSOCHS is considered an optimal solution for environments where sensors can communicate with each other using normal traffic patterns; however, its performance degrades under higher values of power, energy consumption, and computational complexity. In Wang et al. (2019) [33], an even distribution strategy for network resource allocation is presented. It ensures that the amount of data spread across the WSN is distributed evenly across all locations.

Wang et al. (2018) [34] proposed a cluster election algorithm based on PSO. They calculate the amount of power allocated by evaluating the distribution of the energy levels of the sensor nodes and the energy of the cluster heads in the coverage area. Finally, choose the node with the highest energy.

Mehra et al. (2020) [42] proposed a fuzzy-based cluster head selection algorithm by choosing the best candidate for cluster head to provide load balancing. Therefore, in order to solve this problem, it uses the remaining energy, the distance of the sink point, and the density of the node as the input parameters of the fuzzy inference system. In other words, compared with the previous model mentioned in the document, the advantage of this model is that it is more efficient, and the network life is longer than before. Because this method attempts to reduce the data transmission between nodes by selecting cluster head nodes more efficiently than other methods, it has been mentioned in the document. On the other hand, the weakness of this system is that it largely depends on the distance of the nodes.

Fuzzy logic is a decision-making system in most aspects. It has been widely used in knowledge-based systems and multiobjective optimization techniques. Traditional control mechanisms require precise and concise information about their environment. However, fuzzy logic systems can make decisions based on many environmental parameters. Wang et al. (2019) [26] propose a fuzzy system and multiobjective technology based on predefined rules. Wireless sensor networks usually overcome the uncertainty problem by incorporating fuzzy logic into clustering algorithms. In the Fuzzy Clustering Algorithm (FCA) [47], fuzzy logic combines multiple parameters to determine a cluster head for WSNs. FCA uses the IF THEN rule for better precision when using the defuzzification output method. If a node has a strong coupling with its neighbors, it will become the cluster's head. Fuzzy logic technology can be used in distributed and centralized systems.

Besides, positioning services and temporal remote sensing images are crucial to many countries for intelligent agriculture. For example, Brazil has developed functions in the Global Navigation Satellite System (GNSS) multi-constellation data processing and analysis, which can be used for various applications such as agriculture [44]. Another example is the New Zealand government. They allocated funds to support the Regional Satellite Augmentation System (SBAS) [45]. This issue will significantly improve the accuracy of the WSN GPS and apply it to intelligent farming and cadastral mapping purposes.

Paper	Main Context	Limitation		
Wang et al. [33]	Strategy for network	CHs depend on the		
	resource allocation	energy center		
Wang et al. [34]	A cluster election	Not considered the		
	algorithm based on PSO	effectiveness in the		
		dying nodes		
Mehra et al.	A fuzzy-based cluster head	largely depends on the		
[42]	selection	distance of the nodes		
Wang et al. [26]	A fuzzy system and	Based on predefined		
	multiobjective technology	rules		
V. Godbole [47]	Fuzzy logic to determine a	a Limit in residual energy		
	cluster head	and lifespan of the		
		network in large scale		
		network		
Srinivasa et al.	A PSO-based Power	Higher values of power,		
[32]	Saving Cluster Head	energy consumption,		
	Selection	and computational		
		complexity		
Abdul Latiff et	PSO-C based optimal	Not consider the cluster		
al. [31]	cluster head sections	head distance with a		
		base station		

Table 1: Comparison of the existing studies

3 METHODOLOGY

Several studies on energy consumption, resource scarcity, and global warming are now being conducted throughout the world. WSN and other supporting technologies can significantly improve the efficiency of agriculture. It demonstrates that optimal resource utilization is critical in agriculture. A helpful cluster-head selection approach might be utilized to collect data from sensor nodes, improve energy efficiency, and reduce channel congestion to preserve the stability of WSNs in the clustering aspect. As a result, it is critical to extend the life of sensor networks and respond to the



Fig. 2. Overview of a fuzzy system and its features

demands of precision agriculture, which necessitates improved methodologies and technologies to reduce costs and enhance output.

In this research, we use a fuzzy inference system to improve the performance of the previously proposed algorithm and propose an optimal cluster head selection algorithm by reducing memory and communication overhead. The proposed clustering algorithm is based on the functionality of the LEACH protocol. The LEACH protocol is one of the first and most popular layer protocols for selecting WSN cluster leaders. This protocol can divide network activity into four phases (Fig 2). At the beginning of each cycle, the resulting number of nodes is chosen as the cluster head in random order. For this purpose, each node generates a random number between 0 and 1 using Eq. 1. If the generated number is less than T(n), the node is selected as cluster-head. In Eq. 1, P represents the correlation between the number of nodes and clusters in the network. Here, r represents the current round, and G is the number of nodes in the previously selected cluster head.

$$T(n) = \begin{cases} \frac{P}{1 - P \times (r \mod \frac{1}{p})}, & \text{if } n \in G\\ 0, & Otherwise \end{cases}$$
(1)

After the cluster head node is defined, other nodes can be classified according to the strength of the received signal. It helps design a strategy so that each cluster leader can decide to lead a cluster. The cluster head node uses the TDMA mechanism among the cluster members to divide its scope of responsibility for various time slots. In each time segment, a cluster leader communicates with a single cluster member and then shares their packets of information. Each time slot, the cluster leader sends cluster information to the master cluster node. According to this mechanism, cluster headers are modified to handle load distribution among different nodes. So, the algorithm works in the following four stages. Notification phase (advertising), cluster formation phase, forming phase schedule, and data transfer phase.

A. Advertising stage (Notification phase)

Each sensor has chosen a random value between 0 and 1. If the random number is less than the threshold T(n), the node selects the cluster head by itself. This issue is indicated by the

symbol P, representing the percentage of the required cluster head. It is represented by the symbol r, which indicates the number of cycles in the current period. G represents the percent of nodes cluster heads in 1/P of the last recording period. This threshold will determine that each node in period 1/P is a cluster head. With probability P, any node can be the head of the cluster in the first round. The next step increases the probability of selecting a node that is not chosen as the cluster head. The fuzzy inference system (FIS) is described in this study. A broadcast message is sent from each node selected as a cluster head node to other sensors. This message is sent with residual energy from each cluster head node.

ADV = Node ID + Detectable header

In the next step, the non-cluster head node selects the cluster head to which it belongs. The non-cluster head node will join the partner/tightly coupled cluster head if it receives multiple advertisement messages. The strength and correlation of the sensors are determined by the strength of the signals received by the cluster.

B. Cluster formation stage (phase)

After determination, each node must deliver the following information to the cluster head in this phase. When each node decides to join one of the clusters, it sends a join-request message (Join-REQ) back to the cluster head. The messages contain the cluster head's ID, the node's ID, and the header. Each node transmits a join-request message (Join-REQ) back to the chosen cluster head using a CSMA MAC protocol.

$$Join-REQ = Node ID + Cluster-head ID + Header (2)$$

C. Forming stage schedule

The cluster head receives all data from the members and forms a TDMA program to coordinate data transmission within the cluster, and the TDMA prevents head nodes from colliding. The CH node broadcasts an advertisement message, and each non-cluster head node determines a cluster head node. Therefore, each non-cluster head node broadcasts a join request message (JoinREQ) to its cluster head.

D. Data transfer stage

In the data transfer phase, the data transfer begins. If the radio equipment on a non-cluster head node is interrupted, the system has the authority to turn it off. A fuzzy system calculates the number of opportunities (step messages) from each node to the parent node. Based on the "Mamdani" approach, we used a fuzzy system in our case [41]. This system has three input functions and a single output function. The fuzzy value is converted into a crisp value in the output defuzzification operation. The study used a defuzzification method called the central method. One of the more popular methods among defuzzification studies. This method is calculated using the following equation:

Centroid Method =
$$\frac{\int x_i \cdot \mu_i(x_i) dx}{\int \mu_i(x_i) dx}$$
 (3)

The purpose of defuzzification is to put a numerical value on the fuzzy set resulting from the aggregation to be used by the designer (i.e., the tracking step x). Defuzzification can be performed using at least seven operators. Still, the most popular is the centroid, which is calculated by the above formula, for which $\mu(x)$ is the degree of membership of the aggregated fuzzy set for the output x.

An overview of the fuzzy system and its properties is shown in Fig 2. The membership functions (MF) used for our fuzzy system have been tested and optimized, i.e., Gaussian MF. The Gauss M.F. equation is as follows:

Gaussian
$$MF(x; c, \sigma) = e^{-\frac{1}{2}\left(\frac{x-c}{\sigma}\right)^2}$$
 (4)

Where *c* determines the center and σ represents the width of the Gaussian MF, Inputs and outputs are defined as follows: **The first input (residual energy):** Indicates the remaining energy of each node. The higher the energy, the more information can be sent and received. This increases the battery life of the node.

The second input (neighbor's #): It displays the count of each neighboring node. A node with many neighbors is more likely to be a parent node. A proposed method for calculating the number of neighborhoods is given in [28]. This method uses the following formula:

$$N = 1 - e^{-\gamma \pi R^2} \tag{5}$$

Among them, *R* is the radius of the neighbor, $\gamma = 0.01$, so the radius of the operating neighborhood of the algorithm is 15 meters.

The third input (centrality): Is the distance from the central node to the node.

The output (chance): It uses fuzzy rules to combine three input values. The input and output membership functions are shown in Fig. 3.

Input values are defined by fuzzy rules combined to produce an output. The fuzzy system is expressed through the "fuzzy rule base" in Table 2. The relation between the three inputs and the output, which calculates the decision, is as follows:

$$Ds = \sum_{k=1}^{n} E_r N \#(C) \tag{6}$$

Where three inputs are Er, N#, and C, in this equation, E_r denotes residual energy, N# denotes neighbors, and C denotes centrality. Ds represents the output, which is the selection decision.

Proposed Fuzzy-Logic based cluster-head selection algorithm				
Input← Non-cluste	r head WSN			
Output← Cluster-h	ead WSN			
Cluster Head ← em	pty			
# All parameters ar	e included			
# In the beginning,	nodes are chosen as the cluster head in random order			
Start Advertising st	age			
R← Ran	d (0,1)			
If (R <t(n< td=""><th> Then the node selected as Cluster Head </th></t(n<>	 Then the node selected as Cluster Head 			
# Each r	ode selected as CH with P probability			
broadca	st message ()			
Start Cluster forma	ition stage			
Each no	de joined to a cluster			

Each Node join-request message (Join-REQ) to cluster head Start Forming stage schedule

Cluster Head (X) \leftarrow Data of Cluster nodes

Cluster Head broadcast advertising

Start Data transfer stage

data transfer begins If (Radio equipment non-cluster== False) Node turn off

Else

Data transfer continue

the fuzzy system calculates the number of step messages (Nodes

to CH.) **End**



Fig. 3. (a), (b), and (c) the input belonging functions, (d) the belonging output function of the fuzzy system



Fig. 4. The combination of input parameters

Figs 4 and 5 show a combination of the inputs to produce an output.

Rules	Inputs (if)	Output	
		(then)	
1	(Er==Less) && (N# ==few) && (C==far)	smallest	
2	(Er=medium) && (N# ==few) && (C==far)	smallest	
3	(Er=Less) &&(N# ==medium) && (C==far)	smallest	
4	(Er==Less) &&(N#==few) && (C==medium)	smallest	
5	(Er==Less) &&(N#==medium) &&	small	
	(C==medium)	Sinali	
6	(Er==medium) && (N# ==medium) &&	11	
	(C==far)	small	
7	(Er==high) && (N# ==many)	langest	
	&& (C==close)	largest	
8	(Er==high) && (N# ==medium)	1	
	&&(C==close)	large	
9	(Er==medium) && (N# ==many)	large	
	&&(C==close)	large	
10	(Er==high) && (N# ==many)	large	
	&&(C==medium)	large	
11	(Er=medium) && (N# ==medium)	medium	
	&&(C==medium)	meanum	
12	(Er==medium) && (N# ==many)	medium	
	&&(C==medium)	meanum	
13	(Er==medium) && (N# ==medium)	medium	
	&&(C==close)	meanum	
14	(Er==medium) &&(N#==medium) &&(C==far)	medium	
15	(Er==medium) && (N#==few)	medium	
	&&(C==medium)	mearum	
16	(Er=medium) &&(N#=many)	medium	
	&& (C is medium)	meanum	
17	(Er=Less) &&(N#==medium)	medium	
	&& (C==medium)	meanum	
18	(Er==high) && (N# ==medium)		
	&&(C==medium)	medium	

TABLE 2 Fuzzy Inference Rule Base



Fig. 5. Combination of inputs to produce outputs

We considered a network of 100 randomly distributed nodes to test the method. Other network parameters modified in the algorithm are shown below:

TABLE 5 Farameters mitia	
% Maximum value of fields - x ar	nd y maximum (in meters)
Xm	100m
Ym	100m
%Number of Nodes in the field	
Ν	100
%x and y Coordinates of the sink	
Sink.x	0.5*xm
Sink.y	0.5*ym
%The probability that the initial	selection of a node becomes the head
of the cluster	
Р	0.1
%Energy Model (all values in Jou	iles)
%Initial Energy:	
Eo	0.5j
%Eelec=Etx=Erx	
Etx	50*0.000000001
Erx	50*0.000000001
%Data Aggregation Energy	
EDA	5*0.00000001
%Transmit Amplifier types	
Efs	10*0.000000000001
Emp	0.0013*0.00000000001

TABLE 3 Parameters Initialization

3.1 Simulation setup

Simulation parameters used in the proposed method's experimental analysis are presented in this section. Network Simulator2 (NS2) was used in the simulation experiments, which is a well-known and reliable simulation tool for analyzing network routing and communication. Table 3 shows the default values for the simulation parameters. The dimension of the network extends to an area of 100 m \times 100 m, and the coordinates of the base station are [50.50]. There are different numbers of rounds in which the simulation results are evaluated. The period of a single simulation round is set to 1000. Additionally, the number of agriculture sensors is set to 100, respectively. All of the agricultural sensors, i.e., temperature sensors, light sensors, soil moisture sensors, location sensors, airflow sensors, etc. are scattered randomly. Packet size (k) and payload size are set to 64 bits and 256 bytes.

The energy required to run a running transmitter of agriculture sensors is uniform, i.e., 50nj. A Constant Bit Rate (CBR) is used between sensor nodes, and the sensor nodes' transmission range is set to 20m.

4 RESULTS AND DISCUSSION

The features of the provided system model are explained. First, we announce the assumptions of the network model. To confirm the design method, we use a randomization form to consider a test platform network consisting of 100 nodes. The dimension of the network extends to an area of $100 \text{ m} \times 100 \text{ m}$. In this case, the base station coordinates are considered [50.50], and the probability value is set at 0.1. The initial energy is set to 0.5 joules. All sensor nodes are considered to have the same energy state. At the beginning of Cluster head selection, all nodes have the same probability of being a CH. The probability that an initial selection of a node will become a header is determined by P. The energy model of nodes in Joules is determined. Regarding Pseudocode in the previous chapter, the simulation continues until the radio equipment of normal node(s) goes down. So, the algorithm has the propriety of turning the node off and continuing to transfer data by other nodes.

In addition, all nodes and the sink are located in a fixed position with similar computational and memory capacity. The network is considered in a homogenous form. The distance between nodes is calculated as the intensity of the received signal and means that the sensor node does not recognize its exact position. The following is a description of the radio energy model.

The model of K-bit packet transmission radio energy is modeled over distance as follows:

$$E_{kk}(k,d) = \begin{cases} k * E_{elcc} + k * {}^{\mathcal{E}_{fx}} * d^2, d < d_0 \\ k * E_{elcc} + k * {}^{\mathcal{E}_{mp}} * d^4, d < d_0 \end{cases}$$
(7)

The radio energy model receives a K-bit package which can be represented as

$$E_{kk} = E_{elec} \times k \tag{8}$$

 E_{elec} considers the energy required to run a running transmitter circuit depending on the distance. The radio parameters used in the simulation are defined as follows:

Tal	ble	5	Radio	parameters	in	the	simu	lation

E _{elec}	50nJ/bit
EDA	5nJ/bit/report
ϵ_{f}	10pJ/bit/m ²
ϵ_{mp}	0.0013pJ/bit/m ⁴

Algorithms are tested for other values of the case. After about 1000 runs, optimal cluster head values can be obtained. Fig. 6 shows the changes in the system parameters during iteration.



Fig. 6. The difference in system values by changing parameters (Distance)

Fig. 7 shows the resulting output of this system applied at 1000 rounds for 100 nodes with a random distribution. Table 4 illustrates the four-run simulations to compare the lifetime of our algorithm and LEACH.





Fig. 8. (a) Remaining energy, (b) Number of active nodes (Proposed method)

The results show that the precision is higher than the traditional LEACH algorithm. Fig. 8 shows the performance analysis of our system in terms of remaining energy and the number of active nodes. Fig. 9 illustrates the performance comparison between the standard LEACH algorithm and the fuzzy inference system. Part (a) represents the remaining energy, and part (b) represents the number of active nodes. As shown in Fig. 9, using a fuzzy inference system produces better results.



Fig. 9. (a) Remaining energy (b) Number of live nodes

Also, in addition to the LEACH method, it is possible to compare the performance of the proposed method with the other methods. The performance of the proposed method is compared with the FCA method. Given that the FCA method is consistent with the proposed method and also uses fuzzy logic that can show the efficiency of the proposed method by comparison.

The FCA algorithm is designed for WSN with fixed sensor nodes. It adjusts the radius of the radio cluster head according to the remaining energy. It also adjusts the distance between the sensor nodes and the base station. This method aims to select the radius of the cluster head. In this algorithm, a number between 0 and 1 is generated for each node in each round. If the threshold (T(n)) is greater than a certain value, the node can be considered a cluster head. The fuzzy system considers three inputs. The results of this study were simulated. FCA is regarded as a stable and energy-efficient WSN clustering algorithm. Due to the similarity between our proposed method and the LEACH algorithm, we compared it with the LEACH algorithm. Fig. 10 shows the performance comparison between the FCA method and the LEACH algorithm. The results demonstrate that FCA has improved performance over the LEACH. Finally, in Fig. 11, it can be seen that the results of the comparison demonstrate the superiority and improvement of our proposed method. From the simulation results, it is evident that the proposed method results in a 24% improvement in the ratio of alive nodes compared to other methods. Fig. 12 illustrates the amount of residual energy algorithm in which the proposed method is more efficient than other methods, and the lifespan of the network is extended. Simulation results however indicate a 9% improvement in the residual energy ratio of the proposed method.



Fig. 10. Remaining energy comparison in terms of lifetime and residual energy



Fig. 11: The comparison of LEACH and FCA algorithm and the proposed method (No. of live nodes)



Fig. 12. Residual energy comparison between LEACH, FCA, and the proposed algorithm

5 CONCLUSION

We concluded that the WSNs constitute an essential contribution to remote sensing of smart agriculture systems for monitoring, temperature measurement, irrigation system measurement, and water supply measurement, among other things. It plays a vital role in enabling communication between several intelligent agriculture ecosystem computing components. WSN consists of a network of nodes, which communicate with each other and a base station. The sensors have several limitations, including topology management, mapping, and organization, battery power. These challenges adversely affect the intelligent agriculture system's performance. By improving the quality of WSN, data can be accessed remotely. This will enhance the connectivity and coverage area of the farm. An algorithm for fuzzy-based intelligent agricultural system cluster head selection is proposed to overcome the difficulties of unreasonable cluster head selection and excessive power consumption. An extensive evaluation and discussion of other related technologies in this field were conducted, followed by a brief comparison. By comparing this work with existing solutions, we conclude that the proposed method is more efficient than the existing methods. Energy consumption caused improving network lifetime. This study also concluded that the proposed algorithm could apply in innovative agriculture applications to analyze and use a simulation environment.

Abbreviation

WSN	Wireless Sensor Networks
IoT	Internet of Things
TDMA	Time Division Multiple Access
PSO	Particles Swarms Optimizations
FCA	Fuzzy Cluster Algorithm
GNSS	Global Navigation Satellite System
FIS	Fuzzy Inference System
СН	Cluster Head
MF	Membership Function
NS2	Network Simulator (Version 2)
SBAS	Satellite-Based Augmentation System

Declarations

Funding

Not Applicable.

Conflicts of interest/Competing interests

The authors are not affiliated with any organization having a direct or indirect financial interest in the subject dealt with in the manuscript.

Availability of data and materials

If necessary, the document will be provided.

Code availability

If necessary, the code will be provided.

Authors' contributions

All authors contributed to the idea and design of the study. The authors prepared the documents and collected and analyzed the data. Mahdi Safaei wrote the first draft of the manuscript, and all authors commented on earlier versions of the manuscript. The final manuscript was read and approved by all authors.

Ethics approval

All authors hereby ensure that the following contents of the manuscript are accurate:

- This material is the original work of the authors, never published elsewhere.
- This article is not currently being considered for publication elsewhere.
- This article truly and fully reflects the author's own research and analysis.
- This article duly acknowledges the significant contributions of co-authors and lead researchers.
- All sources used are properly disclosed (accurate citations).
- All authors have personally and actively participated in important work on the preparation of articles and are responsible to society for the content of their articles.

Consent to participate

Informed consent was obtained from all individual participants included in the study.

Consent for publication

Participants' informed consent to publish their data and photos.

REFERENCES

1. Kamilaris, Andreas, and Francesc X. Prenafeta-Boldú. "Deep learning in agriculture: A survey." Computers and electronics in agriculture 147 (2018):70-90,Elsevier.DOI: https://doi.org/10.1016/j.compag.2018.02.016.

- Glaroudis, Dimitrios, Athanasios Lossifides, and Periklis Chatzimisios. 2 "Survey, comparison and research challenges of IoT application protocols for smart farming." *Computer Networks* 168 (2020): 107037, Elsevier. DOI: <u>https://doi.org/10.1016/j.comnet.2019.107037</u>
- 3. Devi, V. Seedha, T. Ravi, and S. Baghavathi Priya. "Cluster-Based Data Aggregation Scheme for Latency and Packet Loss Reduction in WSN." Computer Communications 149 (2020): 36-43, Elsevier. DOI: https://doi.org/10.1016/j.comcom.2019.10.003
- Rajpoot, Prince, Shubham Harsh Singh, Rashmi Verma, Kumkum Dubey, 4. Shivendra Kumar Pandey, and Sharad Verma. "Multi-factor-Based Energy-Efficient Clustering and Routing Algorithm for WSN." In Soft Computing: Theories and Applications, pp. 571-581. Springer, Singapore, 2020. DOI: https://doi.org/10.1007/978-981-15-0751-9_53
- 5. Dai, Tianwei, and Zhengtao Ding. "Online distributed distance-based outlier clearance approaches for wireless sensor networks." Pervasive and

Mobile Computing (2020): 101-130, Elsevier. DOI: https://doi.org/10.1016/j.pmcj.2020.101130.

- Jamshidi, Mojtaba, Mehdi Esnaashari, Aso Mohammad Darwesh, and Mohammad Reza Meybodi. "Using Time-Location Tags and Watchdog Nodes to Defend Against Node Replication Attack in Mobile Wireless Sensor Networks." *International Journal of Wireless Information Networks* 27, no. 1 (2020): 102-115, Springer. DOI: <u>https://doi.org/10.1007/s10776-019-00469-0</u>
- Huang, Zhenzhen, QiangNiu, Shuo Xiao, and Tianxu Li. "Energy harvesting algorithm considering max flow problem in wireless sensor networks." *Computer Communications* 150 (2020): 626-633, Elsevier. DOI: <u>https://doi.org/10.1016/j.comcom.2019.12.008</u>
- Pal, Amitangshu, and Alireza Jolfaei. "On the Lifetime of Asynchronous Software-Defined Wireless Sensor Networks." *IEEE Internet of Things Journal* (2020). DOI: <u>https://10.1109/JIOT.2020.2964649</u>
- Das, Rupayan, Dinesh Dash, and Mrinal Kanti Sarkar. "HTMS: Fuzzy Based Hierarchical Trust Management Scheme in WSN." *Wireless Personal Communications* (2020): 1-34, Springer. DOI: <u>https://doi.org/10.1007/s11277-020-07092-w</u>
- Rodrigues, Paul, and Jacob John. "Joint trust: an approach for trust-aware routing in WSN." Wireless Networks (2020): 1-16, Springer. <u>https://doi.org/10.1007/s11276-020-02271-w</u>
- Gbadouissa, Jocelyn EdinioZacko, Ado Adamou Abba Ari, Chafiq Titouna, Abdelhak Mourad Gueroui, and Ousmane Thiare. "HGC: HyperGraph based Clustering scheme for power aware wireless sensor networks." *Future Generation Computer Systems* 105 (2020): 175-183, Elsevier. DOI: https://doi.org/10.1016/j.future.2019.11.043
- Tian, Youliang, Zuan Wang, JinboXiong, and Jianfeng Ma. "A Blockchain-Based Secure Key Management Scheme with Trustworthiness in DWSNs." *IEEE Transactions on Industrial Informatics* (2020). DOI: <u>https://10.1109/TII.2020.2965975</u>
- Elappila, Manu, SuchismitaChinara, and DayalRamakrushnaParhi. "Survivability Aware Channel Allocation in WSN for IoT applications." *Pervasive and Mobile Computing* 61 (2020): 101-107, Elsevier. DOI: <u>https://doi.org/10.1016/j.pmcj.2019.101107</u>
- Srikanth, Nandoori, and Muktyala Siva Ganga Prasad. "A Compressive Family Based Efficient Trust Routing Protocol (C-FETRP) for Maximizing the Lifetime of WSN." In *Data Communication and Networks*, pp. 69-80. Springer, Singapore, 2020. DOI: <u>https://doi.org/10.1007/978-981-15-0132-6_6</u>
- Kumaresan, K., and S. Nithya Kalyani. "Energy efficient cluster based multilevel hierarchical routing for multi-hop wireless sensor network." *Journal of Ambient Intelligence and Humanized Computing* (2020): 1-10, Springer. DOI: <u>https://doi.org/10.1007/s12652-020-01700-0</u>
- Malakar, Madhuri. "TLBO Based Cluster-Head Selection for Multiobjective Optimization in Wireless Sensor Networks." In *Nature Inspired Computing for Wireless Sensor Networks*, pp. 303-319. Springer, Singapore, 2020. DOI: <u>https://doi.org/10.1007/978-981-15-2125-6_13</u>
- Nehra, Vibha, Ajay K. Sharma, and Rajiv K. Tripathi. "FIEPE: Fuzzy Inspired Energy Efficient Protocol for Heterogeneous Wireless Sensor Network." *Wireless Personal Communications* 110, no. 4 (2020): 1769-1794, Springer. DOI: https://doi.org/10.1007/s11277-019-06811-2
- Thiagarajan, R. "Energy consumption and network connectivity based on Novel-LEACH-POS protocol networks." *Computer Communications* 149 (2020): 90-98, elsevier. DOI: <u>https://doi.org/10.1016/j.comcom.2019.10.006</u>
- Shukla, Anurag, and Sarsij Tripathi. "An Effective Relay Node Selection Technique for Energy Efficient WSN-Assisted IoT." *Wireless Personal Communications* (2020): 1-31, Springer. DOI: <u>https://doi.org/10.1007/s11277-020-07167-8</u>
- Raj, PV Pravija, Ahmed M. Khedr, and Zaher Al Aghbari. "Data gathering via mobile sink in WSNs using game theory and enhanced ant colony optimization." *Wireless Networks* (2020): 1-16, Springer. DOI: https://doi.org/10.1007/s11276-020-02254-x
- Arora, Vishal Kumar, Vishal Sharma, and Monika Sachdeva. "A multiple pheromone ant colony optimization scheme for energy-efficient wireless sensor networks." *Soft Computing* 24, no. 1 (2020): 543-553, Springer. DOI: <u>https://doi.org/10.1007/s00500-019-03933-4</u>
- Ngangbam, Remika, Ashraf Hossain, and Alok Shukla. "Improved low energy adaptive clustering hierarchy and its optimum cluster head selection." *International Journal of Electronics* 107, no. 3 (2020): 390-402, Taylor and Francis Online. DOI: https://doi.org/10.1080/00207217.2019.1661023
- Han, Tao, Lijuan Zhang, Sandeep Pirbhulal, Wanqing Wu, and Victor Hugo C. de Albuquerque. "A novel cluster head selection technique for

edge-computing based IoMT systems." *Computer Networks* 158 (2019): 114-122, Elsevier. DOI: https://doi.org/10.1016/j.comnet.2019.04.021

- Shalini, V. Baby, and V. Vasudevan. "Achieving energy efficient wireless sensor network by choosing effective cluster head." *Cluster Computing* 22, no. 4 (2019): 7761-7768, Springer. DOI: https://doi.org/10.1007/s10586-017-1375-5
- El Gheche, Mireille, Giovanni Chierchia, and Pascal Frossard. "OrthoNet: Multilayer Network Data Clustering." *IEEE Transactions on Signal and Information Processing over Networks* 6 (2020): 13-23. DOI: https://doi.org/10.1109/TSIPN.2020.2970313
- Lei Du, Zhihua Cui, Xingjuan Cai, and Liping Xie. "A Multiobjective Localization Algorithm with Real Average Distance in WSN." In 2019 IEEE 21st International Conference on High-Performance Computing and Communications(HPCC), pp. 29-36. DOI: https://10.1109/HPCC/SmartCity/DSS.2019.00019.
- Balaji S, Julie EG, Robinson YH. Development of fuzzy based energy efficient cluster routing protocol to increase the lifetime of wireless sensor networks. *Mobile Networks and Applications*. 2019 Apr 15;24(2):394-406, Springer. DOI: <u>https://doi.org/10.1007/s11036-017-0913-y</u>
- Biswas, Tanmay, Sushil Kumar, Tapaswini Singh, Kapil Gupta, and Deepika Saxena. "A comparative Analysis of Unequal Clustering-Based Routing Protocol in WSNs." In *Soft Computing and Signal Processing*, pp. 53-62. Springer, Singapore, 2019. DOI: <u>https://doi.org/10.1007/978-981-13-3600-3_5</u>
- Heinzelman,W.R.; Chandrakasan, A.; Balakrishnan, H. Energy-efficient communication protocol for wireless micro-sensor networks. In *Proceedings of the 33rd Annual Hawaii International Conference on System Sciences*, Maui, HI, USA, 4–7 January 2000; pp. 1–10, IEEE. DOI: <u>https://doi.org/10.1109/HICSS.2000.926982</u>.
- Heinzelman, W.; Chandrakasan, A.; Balakrishnan, H. An Application-Specific Protocol Architecture for Wireless Microsensor Networks. *IEEE Transactions on Wireless Communications*, 2002, 1, 660–670. DOI: https://doi.org/10.1109/TWC.2002.804190.
- Abdul Latiff, N.M.; Tsimenidis, C.C.; Sharif, B.S. Energy-Aware Clustering for wireless Sensor Networks using Particle Swarm Optimization. In Proceedings of the *IEEE 18th International Symposium* on Personal, Indoor and Mobile Radio Communications, Athens, Greece, 3–7 September 2007; pp. 1–5. DOI: <u>https://doi.org/10.1109/PIMRC.2007.4394521</u>.
- Srinivasa Rao, P.C.; Prasanta, K.; Jana Haider, B. A particle swarm optimization based energy efficient clusterhead selection algorithm for wireless sensor networks, *Wireless Networks*, 2017, 23, 2005–2020, Springer. DOI: <u>https://doi.org/10.1007/s11276-016-1270-7</u>
- 33. Wang, Jin, Yu Gao, Wei Liu, Arun Kumar Sangaiah, and Hye-Jin Kim. "An improved routing schema with special clustering using PSO algorithm for heterogeneous wireless sensor network." *Sensors*, 19, no. 3 (2019): 671, MDPI. DOI: https://doi.org/10.3390/s19030671
- Wang, Jin, Chunwei Ju, Yu Gao, Arun Kumar Sangaiah, and Gwang-jun Kim. "A PSO based energy efficient coverage control algorithm for wireless sensor networks." *Computers, Materials and Continua*, 56, no. 3 (2018): 433-446, Tech Science Press. DOI: https://doi.org/10.3970/cmc.2018.04132
- Colomina, Ismael, and Pere Molina. "Unmanned aerial systems for photogrammetry and remote sensing: A review." *ISPRS Journal of photogrammetry and remote sensing* 92 (2014): 79-97, Elsevier. DOI: <u>https://doi.org/10.1016/j.isprsjprs.2014.02.013</u>
- Barrientos, Antonio, Julian Colorado, Jaime del Cerro, Alexander Martinez, Claudio Rossi, David Sanz, and Joao Valente. "Aerial remote sensing in agriculture: A practical approach to area coverage and path planning for fleets of mini aerial robots." *Journal of Field Robotics* 28, no. 5 (2011): 667-689, Wiley. DOI: <u>https://doi.org/10.1002/rob.20403</u>
- Turner, Darren, Arko Lucieer, and Christopher Watson. "An automated technique for generating geo-rectified mosaics from ultra-high resolution unmanned aerial vehicle (UAV) imagery, based on structure from motion (SfM) point clouds." *Remote sensing* 4, no. 5 (2012): 1392-1410, MDPI. DOI: <u>https://doi.org/10.3390/rs4051392</u>
- Tomlinson, Isobel. "Doubling food production to feed the 9 billion: a critical perspective on a key discourse of food security in the UK" *Journal of rural studies* 29 (2013): 81-90, Elsevier. DOI: https://doi.org/10.1016/j.jrurstud.2011.09.001
- Al-qaness, Mohammed AA, Mohamed Abd Elaziz, Sunghwan Kim, Ahmed A. Ewees, Aaqif Afzaal Abbasi, Yousif A. Alhaj, and Ammar Hawbani. "Channel state information from pure communication to sense

and track human motion: A survey." Sensors 19, no. 15 (2019): 3329, MDPI. DOI: https://doi.org/10.3390/s19153329

- 40. Gupta, Varsha, and Shashi Kumar Sharma. "Cluster head selection using modified ACO." In Proceedings of fourth international conference on soft computing for problem solving, pp. 11-20. Springer, New Delhi, 2015. DOI: https://doi.org/10.1007/978-81-322-2217-0 2
- 41. M. Negnevitsky, Artificial intelligence: A guide to intelligent systems, Addison-Wesley, Reading, MA, 2001.
- 42. Pawan Singh Mehra, Mohammad Najmud Doja, and Bashir Alam. "Fuzzy based enhanced cluster head selection (FBECS) for WSN" Journal of King Saud University - Science 32 (2020): 390-401, Elsevier. DOI: https://doi.org/10.1016/j.jksus.2018.04.031
- 43. Felix Mazunga and Action Nechibvute. "Ultra-low power techniques in energy harvesting wireless sensor networks: Recent advances and issues" Scientific African no. 11 (2021): 1-14, Elsevier. https://doi.org/10.1016/j.sciaf.2021.e00720
- 44. https://www.gsc-europa.eu/news/calibra-bringing-precision-agricultureto-brazil-3
- 45. https://www.linz.govt.nz/data/geodetic-services/satellite-basedaugmentation-system
- Gheisari Mehdi, Afzaal Abbasi Aaqif, Sayari Zahra, Rizvi Qasim, 46. Asheralieva Alia, Banu Sabitha, M. Awaysheh Feras, Hussain Shah Syed Bilal, Arif Raza Khuhawar. "A Survey on Clustering Algorithms in Wireless Sensor Networks: Challenges, Research, and Trends" 2020 International Computer Symposium (ICS), IEEE (2020): https://doi.org/10.1109/ICS51289.2020.00065
- 47. V. Godbole, FCA An Approach On LEACH Protocol Of Wireless Sensor Networks Using Fuzzy Logic, International Journal of Computer Communications and Networks (IJCCN), 2(3) (2012), 1-13. http://doi.org/10.48550/arXiv.1301.1753
- 48. Kumar, S.A.; Ilango, P. The Impact of Wireless Sensor Network in the Field of Precision Agriculture: A Review. Wirel. Pers. Commun. 2018, 98, 685-698. https://doi.org/10.1007/s11277-017-4890-z
- 49. Wu, F.; Geng, Y.; Tian, X.; Zhong, S.; Wu, W.; Yu, S.; Xiao, S. Responding Climate Change: A Bibliometric Review on Urban Environmental Governance. J. Clean. Prod. 2018, 204, 344-354. https://doi.org/10.1016/j.jclepro.2018.09.067
 - 50. [1] Gheisari, M., Esnaashari, M. (2017). A

survey to face recognition algorithms:

advantageous and

disadvantageous. Journal Modern

Technology & Engineering, V. 2(1), pp. 57-65.

- 51. @Article{imagesurvey,
- 52. author = {M.Gheisari and M. Esnaashari},
- 53. title = {A survey to face recognition algorithms: advantageous and disadvantageous},
- 54. journal = {Journal Modern Technology & Engineering},
- 55. year = $\{2017\}$,
- 56. volume = $\{2\}$,
- 57. number = $\{1\}$,
- 58. pages = $\{57-65\}$,
- 59. }

```
60.
```

61. [2] Gheisari, M., Baloochi, H., Gharghi, M., & Khajehyousefi, M. (2012). An Evaluation of Two Proposed Systems of

Sensor Data's Storage in Total Data Parameter. International Geoinformatics Research and Development Journal.

- 62. @article{gheisari2012evalu ation,
- 63. title={An Evaluation of Two Proposed Systems of Sensor Data's Storage in Total Data Parameter},
- 64. author={Gheisari, Mehdi and Baloochi, Hamed and Gharghi, Meysam and Khajehyousefi, Mehdi},
- 65. journal={International Geoinformatics Research and Development Journal },
- 66. year={2012} }

```
67.
```

68.

69.

- 70. [3] Gheisari, M.. Design, Implementation and Evaluation of SemHD: a New Semantic Hierarchical Sensor Data Storage. Indian Journal of **Innovations and Developments**, [S.l.], p. 115-120, mar. 2012. ISSN 2277 - 5390. Available at: <http://ijid.informaticspublishing.com/in dex.php/ijid/article/view/31604>. Date accessed: 21 Jun. 2018.
- 71.
- 72. @Article{SemHD,
- 73. author = $\{GHEISARI, M\}$,
- 74. title = {Design, Implementation and Evaluation of SemHD},
- 75. journal = {Indian Journal of Innovations and Developments},

```
76.
           year =
                        \{2012\},\
77.
           pages =
                        \{115-120\},\
78.
           month =
                        {March}
79.
         }
80.
```

81.[4] M. Gheisari et al., "NSSSD: A new semantic hierarchical storage for sensor data," 2016 IEEE 20th International Conference on Computer Supported Cooperative Work in Design (CSCWD), Nanchang, 2016, pp. 174-179.

82.

```
83.
      @INPROCEEDINGS{7565984,
   author={M. Gheisari and A. A.
   Movassagh and Y. Qin and J.
   Yong and X. Tao and J. Zhang
   and H. Shen},
   booktitle={2016 IEEE 20th
   International Conference on
   Computer Supported Cooperative
   Work in Design (CSCWD) },
   title={NSSSD: A new semantic
   hierarchical storage for sensor
   data},
   year={2016},
   volume={},
   number={},
   pages = \{174 - 179\},\
   month={May}, }
84.
85.
        [5] M. Gheisari, G. Wang
86.
   and M. Z. A. Bhuiyan, "A Survey
   on Deep Learning in Big
   Data," 2017 IEEE International
   Conference on Computational
   Science and Engineering (CSE) and
   IEEE International Conference on
   Embedded and Ubiquitous
   Computing (EUC), Guangzhou,
   2017, pp. 173-180.
87.
88.
       @INPROCEEDINGS{8005992,
89.
   author={M. Gheisari and G. Wang
   and M. Z. A. Bhuiyan},
   booktitle={2017 IEEE
   International Conference on
   Computational Science and
   Engineering (CSE) and IEEE
   International Conference on
   Embedded and Ubiquitous
   Computing (EUC) },
   title={A Survey on Deep
   Learning in Big Data},
   year = \{2017\},\
   volume={2},
   number={},
   pages = \{173 - 180\},\
   month={July}, }
90.
91. [6] M. Jafari, J. Wang, Y. Qin, M.
   Gheisari, A. S. Shahabi and X. Tao,
   "Automatic text summarization
```

using fuzzy inference," 2016 22nd

International Conference on

Automation and Computing (ICAC), Colchester, 2016, pp. 256-260.

```
92.
```

```
93.
```

```
@INPROCEEDINGS{7604928,
94.
   author={M. Jafari and J. Wang
   and Y. Qin and M. Gheisari and
  A. S. Shahabi and X. Tao},
  booktitle={2016 22nd
   International Conference on
   Automation and Computing
   (ICAC) },
   title={Automatic text
   summarization using fuzzy
  inference},
   year={2016},
   volume={},
  number={},
  pages = \{256 - 260\},\
  ISSN=\{\},\
  month={Sept}, }
95.
```

```
96.
```

97. [6] Gheisari, Mehdi, et al. "An Evaluation of Two Proposed Systems of Sensor Data's Storage in Total Data Parameter." *International Geoinformatics Research and Development Journal* (2012).

98.

```
99.
       @article{gheisari2012evalu
  ation,
100.
         title={An Evaluation of
   Two Proposed Systems of Sensor
  Data's Storage in Total Data
  Parameter},
101.
         author={Gheisari, Mehdi
   and Baloochi, Hamed and
  Gharghi, Meysam and
  Khajehyousefi, Mehdi},
102.
          journal={International
  Geoinformatics Research and
  Development Journal },
103.
         year={2012}
104.
       }
105.
        [7] Gheisari, Mehdi. "The
```

105. [7] Gheisari, Mehdi. "The Effectiveness of Schema Therapy Integrated with Neurological Rehabilitation Methods to Improve Executive Functions in Patients with Chronic Depression." *Health Science Journal* 10.4 (2016) page:1-6.

```
106.
107.
        @article{gheisari2016effec
   tiveness,
108.
          title={The Effectiveness
   of Schema Therapy Integrated
   with Neurological
   Rehabilitation Methods to
   Improve Executive Functions in
   Patients with Chronic
   Depression},
          author={Gheisari,
109.
   Mehdi},
110.
          journal={Health Science
   Journal},
111.
          volume=\{10\},
112.
          number=\{4\},
113.
          year={2016},
114.
        page = \{1:6\}
115.
          publisher={iMedPub}
116.
        }
117.
118.
119.
        [8] Rezaeiye, Payam Porkar, and
   Mehdi Gheisari. "Performance analysis of
   two sensor data storages." Proceedings of
   2nd International Conference on Circuits.
   Systems, Communications & Computers
   (CSCC). 2011.
120.
121.
        @inproceedings{rezaeiye201
   1performance,
122.
         title={Performance
   analysis of two sensor data
   storages},
         author={Rezaeiye, Payam
123.
   Porkar and Gheisari, Mehdi},
          booktitle={Proceedings
124.
   of 2nd International Conference
   on Circuits, Systems,
   Communications \& Computers
   (CSCC) },
125.
          pages = \{133 - -136\},\
126.
          year={2011}
127.
        }
128.
129.
130.
131.
        [9] Gheisari, Mehdi, et al. "A
   Comparison with some Sensor Network
   Storages." International Conference on
   Computer and Computer Intelligence
   (ICCCI 2011). ASME Press, 2011.
132.
        @inproceedings{gheisari201
133.
   1comparison,
134.
         title={A Comparison with
   some Sensor Network Storages},
         author={Gheisari, Mehdi
135.
   and Bahekmat, Maliheh and
```

```
Setoodeh, Hamid Reza and
   Khajehyousefi, Mehdi},
          booktitle={International
136.
   Conference on Computer and
   Computer Intelligence (ICCCI
   2011) },
137.
          year = \{2011\},\
138.
          organization={ASME
   Press}
139.
        }
140.
141.
142.
143.
        [10] GhadakSaz, Ehsan, et al.
   "Design, Implement and Compare two
   proposed sensor data's storages Named
   SemHD and SSW." From Editor in
   Chief (2012): 78.
144.
145.
        @article{ghadaksaz2012desi
   gn,
146.
          title={Design, Implement
   and Compare two proposed sensor
   data's storages Named SemHD and
   SSW},
147.
          author={GhadakSaz, Ehsan
   and Amini, Mohammad Reza and
   Porkar, Payam and Gheisari,
   Mehdi},
148.
          journal={From Editor in
   Chief},
149.
          pages = \{78\},\
150.
          year={2012}
151.
        }
152.
153.
        [11] Rezaeiye, Payam Porkar, et
154.
   al. "Statistical method used for doing
   better corneal junction
   operation." Advanced Materials Research.
   Vol. 548. Trans Tech Publications, 2012.
155.
156.
157.
        @inproceedings{rezaeiye201
   2statistical,
158.
          title={Statistical
   method used for doing better
   corneal junction operation},
159.
          author={P.P.Rezaeiye and
   P.P. Rezaeiye and E.Karbalayi
   and M.Gheisari},
160.
          booktitle={Advanced
   Materials Research },
          volume=\{548\},
161.
162.
          pages = \{762 - 766\},\
163.
         year={2012},
164.
          organization={Trans Tech
   Publ}
165.
166.
        }
```

167. 168. [12] Gheisari, M., A. R. Bagheri. "SHD: a New Sensor Data Storage." In 5th international symposium on advances in science & technology. 2011. 169. 170. @inproceedings{gheisari201 1shd, title={SHD: a New Sensor 171. Data Storage}, 172. author={Gheisari, M and Bagheri, AR }, 173. booktitle={In 5th international symposium on advances in science $\$ technology}, 174. year={2011} 175. } 176. 177. 178. [13] Porkar, Payam, et al. "A Comparison with Two Sensor Data Storages in Energy." International Conference on Computer and Computer Intelligence (ICCCI 2011). ASME Press, 2011. 179. 180. 181. @inproceedings{porkar2011c omparison, 182. title={A Comparison with Two Sensor Data Storagesin Energy}, 183. author={P.P.Porkar and M.Gheisari and G.H.Bazyari and Z.Kaviyanjahromi}, booktitle={International 184. Conference on Computer and Computer Intelligence (ICCCI 2011)}, 185. year={2011}, 186. organization={ASME Press} 187. } 188. 189. [14] Gheisari, Mehdi, and Mehdi Esnaashari. "Data Storages in Wireless Sensor Networks to Deal With Disaster Management." Emergency and Disaster Management: Concepts, Methodologies, Tools, and Applications. IGI Global, 2019. 655-682. 190. @incollection{gheisari2019 191. data, 192. title={Data Storages in Wireless Sensor Networks to Deal With Disaster Management }, author={M.Gheisari and 193. M.Esnaashari},

194. booktitle={Emergency and Disaster Management: Concepts, Methodologies, Tools, and Applications}, $pages = \{655 - -682\},\$ 195. 196. year={2019}, 197. publisher={IGI Global} 198. } 199. 200. 201. 202. [16] Rezaeiye, Payam Porkar, et al. "Agent programming with object oriented (C++)." Electrical, Computer and Communication Technologies (ICECCT), 2017 Second International Conference on. IEEE. 2017. 203. 204. 205. @inproceedings{rezaeiye201 7agent, title={Agent programming 206. with object oriented (C++) }, author={ P.P. Rezaeiye, 207. and p.p. Rezaeiye and E. F. Gh. M. Beig and H. Mohseni and R. Kaviani and M. Gheisari and M. Golzar}, 208. booktitle={Electrical, Computer and Communication Technologies (ICECCT), 2017 Second International Conference on}, 209. $pages = \{1 - -10\},\$ 210. $year = \{2017\},\$ 211. organization={IEEE} 212. } 213. 214. 215. [17] Gheisari, Mehdi, et al. "MAPP: A Modular Arithmetic Algorithm for Privacy Preserving in IoT." Ubiquitous Computing and Communications (ISPA/IUCC), 2017 IEEE International Symposium on Parallel and Distributed Processing with Applications and 2017 IEEE International Conference on. IEEE, 2017. 216. 217. @inproceedings{gheisari201 7mapp, 218. title={MAPP: A Modular Arithmetic Algorithm for Privacy Preserving in IoT}, author={M,Gheisari and 219. G, Wang and M.D.Z.A, Bhuiyan and Z, Wei}, 220. booktitle={Ubiquitous Computing and Communications (ISPA/IUCC), 2017 IEEE

```
International Symposium on
   Parallel and Distributed
   Processing with Applications
   and 2017 IEEE International
   Conference on },
221.
          pages = \{897 - -903\},\
222.
         year = \{2017\},\
223.
          organization={IEEE}
224.
        }
225.
226.
227.
        [18] Ashourian, Mohsen, Mehdi
   Gheisari, and Ali Hashemi. "An Improved
   Node Scheduling Scheme for Resilient
   Packet Ring Network." Majlesi Journal of
   Electrical Engineering 9.2 (2015): 43.
228.
229.
        @article{ashourian2015impr
   oved,
230.
          title={An Improved Node
   Scheduling Scheme for Resilient
   Packet Ring Network },
231.
         author={M,Ashourian and
   M, Gheisari and A, Hashemi},
232. journal={Majlesi Journal
   of Electrical Engineering},
233. volume={9},
234.
         number=\{2\},
235.
        pages={43},
       year={2015},
236.
237.
         publisher={Islamic Azad
   University Majlesi }
238.
      }
239.
240.
241.
        [19] Sharifzadeh, Manaf, Kaveh
   Bashash, Shahram Bashokian, and mehdi
   gheisari. "A Comparison with two
   semantic sensor data storages in total
   data transmission." arXiv preprint
   arXiv:1401.7499 (2014).
242.
243.
       @article{sharifzadeh2014co
   mparison,
244.
       title={A Comparison with
   two semantic sensor data
   storages in total data
   transmission},
245.
         author={Sharifzadeh,
   Manaf and Bashash, Kaveh and
   Bashokian, Shahram and others},
         journal={arXiv preprint
246.
   arXiv:1401.7499},
247.
          year={2014}
248.
        }
249.
250.
251.
        [20] Porkar, P., Mojtaba Fazli,
   and M. Gheisari. "Sensor networks
```

```
challenges." 11th international conference
   on data networks. DNCOCO '12. 2012.
252.
253.
        @inproceedings{porkar2012s
   ensor,
254.
          title={Sensor networks
   challenges},
255.
        author={Porkar, P and
   Fazli, Mojtaba and Gheisari,
   M},
256.
          booktitle={11th
   international conference on
   data networks, DNCOCO '12},
          year={2012}
257.
258.
        }
259.
260.
261.
262.
        [22] Khajehyousefi, Mehdi, et al.
   "A Comparison with Three Proposed
   Sensors Data's Storages." International
   Conference on Advanced Computer
   Theory and Engineering, 4th (ICACTE
   2011). ASME Press, 2011.
263.
264.
        @inproceedings{khajehyouse
   fi2011comparison,
265.
          title={A Comparison with
   Three Proposed Sensors Data's
   Storages},
          author={Khajehyousefi,
266.
   Mehdi and Karimi, Mehdi and
   Bazyari, Gholam Hossein and
   Gheisari, Mehdi},
         booktitle={International
267.
   Conference on Advanced Computer
   Theory and Engineering, 4th
   (ICACTE 2011) },
268.
         year={2011},
269.
          organization={ASME
   Press}
270.
      }
271.
272.
273.
        [23] Fakhimi, Esmaeil, et al.
   "Design Two Sensor Data
   Storages." International Conference on
   Advanced Computer Theory and
   Engineering, 4th (ICACTE 2011). ASME
   Press, 2011.
274.
275.
        @inproceedings{fakhimi2011
   design,
         title={Design Two Sensor
276.
   Data Storages},
         author={Fakhimi, Esmaeil
277.
   and Ajandak, Ghorban and
   Gheisari, Mehdi and Bahekmat,
   Maliheh},
```

278. booktitle={International Conference on Advanced Computer Theory and Engineering, 4th (ICACTE 2011) }, 279. year={2011}, 280. organization={ASME Press} 281. } 282. 283. [24] Gheisari, Mehdi. "Design, implementation and evaluation of SemHD: a new semantic hierarchical sensor data storage." Indian Journal of Innovations and Developments 1.3 (2012): 115-120. 284. @article{gheisari2012desig 285. n, 286. title={Design, implementation and evaluation of SemHD: a new semantic hierarchical sensor data storage}, 287. author={Gheisari, Mehdi}, 288. journal={Indian Journal of Innovations and Developments}, 289. volume={1}, 290. number= $\{3\}$, pages={115--120}, year={2012} 291. 292. 293. } 294. 295 296. [25] Rezaeiye, Payam Porkar, et al. "Creating an ontology using protégé: concepts and taxonomies in brief." 297. 298. @Article{rezaeiyecreating, 299. author = {Rezaeiye, Payam Porkar and Fazli, Mojtaba and Sharifzadeh, Manaf and Moghaddam, Hani and Gheisari, Mehdi}, 300. title = {Creating an ontology using protege: concepts and taxonomies in brief}, journal = {Advances in 301. Mathematical and Computational Methods}, 302. volume={1}, 303. number= $\{3\}$, 304. pages={115--120}, year={2012} 305. 306. } 307. 308. [26] Gheisari M., Wang G., Chen S., Ghorbani H. (2018) IoT-SDNPP: A Method for Privacy-

Preserving in Smart City with Software Defined Networking. In: Vaidya J., Li J. (eds) Algorithms and Architectures for Parallel Processing. ICA3PP 2018. Lecture Notes in Computer Science, vol 11337. Springer, Cham 309. 310. @InProceedings{10.1007/978 -3-030-05063-4 24. 311. author="Gheisari, Mehdi and Wang, Guojun and Chen, Shuhong and Ghorbani, Hamidreza", title="IoT-SDNPP: A Method for Privacy-Preserving in Smart City with Software Defined Networking", 312. booktitle="Algorithms and Architectures for Parallel Processing", 313. year="2018", 314. publisher="Springer International Publishing", 315. pages="303--312", 316. } 317. 318. [28] Mehdi Gheisari, Guojun 319. Wang, Shuhong Chen, Ali Seyfollahi, A Method for Privacy-preserving in IoT-SDN Integration Environment, 16th IEEE International Symposium on Parallel and Distributed Processing with Applications (ISPA 2018), 11-13 Dec. 2018, Melbourne, Australia 320. 321. @InProceedings { ISPA2018, 322. author = {Mehdi Gheisari, Guojun Wang, Shuhong Chen, Ali Seyfollahi}, 323. title = {A Method for Privacy-preserving in IoT-SDN Integration Environment }, 324. booktitle = {16th IEEE International Symposium on Parallel and Distributed Processing with Applications (ISPA 2018)}, 325. year $= \{11-13 \text{ Dec.}\}$ 2018, Melbourne, Australia}, journal = {16th IEEE 326. International Symposium on Parallel and Distributed Processing with Applications (ISPA 2018) }, 327. } 328.

329. [29] M. M. Motahari Kia, J. A. Alzubi, M. Gheisari, X. Zhang, M. Rahimi and Y. Qin, "A Novel Method for Recognition of Persian Alphabet by Using Fuzzy Neural Network," in IEEE Access, vol. 6, pp. 77265-77271, 2018.

330. 331.

@ARTICLE{ACCESSOCR, 332. author={M. M. Motahari Kia and J. A. Alzubi and M. Gheisari and X. Zhang and M. Rahimi and Y. Qin}, journal={IEEE Access}, title={A Novel Method for **Recognition of Persian Alphabet** by Using Fuzzy Neural Network}, year={2018}, volume={6}, number={},

pages={77265-77271},

month={},}

```
333.
```

[30] Alzubi J.A., Yaghoubi A., 334. Gheisari M., Qin Y. (2018) Improve Heteroscedastic Discriminant Analysis by Using CBP Algorithm. In: Vaidya J., Li J. (eds) Algorithms and Architectures for Parallel Processing. ICA3PP 2018. Lecture Notes in Computer Science, vol 11335. Springer, Cham

335.

- @InProceedings{10.1007/978 336. -3-030-05054-2 10,
- author="Alzubi, Jafar 337. A.and Yaghoubi, Ali and Gheisari, Mehdi and Oin, Yongrui",

```
title="Improve
338.
  Heteroscedastic Discriminant
  Analysis by Using CBP
  Algorithm",
```

```
339.
       booktitle="Algorithms and
  Architectures for Parallel
  Processing",
```

```
340. year="2018",
```

```
publisher="Springer
341.
  International Publishing",
```

```
342. address="Cham",
```

```
pages="130--144",
343.
       }
```

```
344.
```

```
345.
```

346. [31] Jayaraman Sethuraman, Jafar Alzubi, Ramachandran Manikandan, Mehdi Gheisari* and Ambeshwar Kumar, "Eccentric Methodology with Optimization to Unearth Hidden Facts of Search Engine Result Pages", Recent Patents on Computer Science (2018) 11:1. https://doi.org/10.2174/2213275911666 181115093050 347. 348. @Article{searchengine, 349. $author = {Jayaraman}$ Sethuraman, Jafar Alzubi, Ramachandran Manikandan, Mehdi Gheisari}, 350. title = {Eccentric Methodology with Optimization to Unearth Hidden Facts of Search Engine Result Pages }, journal = {Recent 351. Patents on Computer Science}, $= \{2018\},\$ 352. year 353. pages $= \{11:1\},\$ 354. ł 355. 356. [32] Mehdi Gheisari, Guojun Wang, Wazir Zada Khan, Christian Fernández-Campusano, 357. A context-aware privacy-preserving method for IoT-based smart city using Software Defined Networking, Computers & Security, Volume 87, 2019, 101470, ISSN 0167-4048, 358. https://doi.org/10.1016/j.cose.2019.02. 006. 359. (http://www.sciencedirect.com/science /article/pii/S0167404818313336) 360. CCF B, IF=3.0 361. 362. @article{GHEISARI2019101470, 363. title = "A context-aware privacypreserving method for IoT-based smart city using Software Defined Networking", 364. journal = "Computers & Security", 365. volume = "87", 366. pages = "101470", 367. year = "2019", 368. issn = "0167-4048", 369. doi = "https://doi.org/10.1016/j.cose.2019.02. 006". 370. url =

"http://www.sciencedirect.com/science/a rticle/pii/S0167404818313336",

```
371. author = "Mehdi Gheisari and Guojun
   Wang and Wazir Zada Khan and
   Christian Fernández-Campusano",
372. }
373.
374.
        [33] Mehdi Gheisari, Guojun
   Wang, Shuhong Chen,
        An Edge Computing-enhanced
375.
   Internet of Things Framework for
   Privacy-preserving in Smart City,
376.
        Computers & Electrical
   Engineering, Volume
   81,2020,106504,ISSN 0045-7906,
377.
378.
        @article{GHEISARI202010650
   4,
379.
        title = "An Edge
   Computing-enhanced Internet of
   Things Framework for Privacy-
   preserving in Smart City",
380. journal = "Computers &
  Electrical Engineering",
381. volume = "81",
       pages = "106504",
382.
      year = "2020",
383.
       issn = "0045-7906",
384.
385.
        doi =
   "https://doi.org/10.1016/j.comp
   eleceng.2019.106504",
386.
       url =
   "http://www.sciencedirect.com/s
   cience/article/pii/S00457906183
   29082",
        author = "Mehdi Gheisari
387.
   and Guojun Wang and Shuhong
   Chen",
388.
        }
389.
390.
        [34] Gheisari, M., Alzubi, J.,
391.
   Zhang, X. et al. Correction to: A new
   algorithm for optimization of quality
   of service in peer to peer wireless
   mesh networks. Wireless
   Netw 25, 4445 (2019).
   https://doi.org/10.1007/s11276-019-
   02016-4
392.
393.
        @Article{Gheisari2019,
394.
        author="Gheisari, Mehdi
395.
        and Alzubi, Jafar
396.
        and Zhang, Xioabo
397.
        and Kose, Utku
398.
        and Saucedo, Jose Antonio
   Marmolejo",
399.
       title="A new algorithm for
   optimization of quality of
```

service in peer to peer wireless mesh networks", 400. journal="Wireless Networks", year="2019", 401. month="Mar", 402. 403. day="23", 404. } 405. 406. [35] X. Zhang, F. Fan, M. Gheisari and G. Srivastava, "A Novel Auto-Focus Method for Image **Processing Using Laser** Triangulation," in IEEE Access, vol. 7, pp. 64837-64843, 2019. doi: 10.1109/ACCESS.2019.2914186 407. 408. 409. @ARTICLE{8703430, author={X. {Zhang} and F. {Fan} and M. {Gheisari} and G. {Srivastava}}, journal={IEEE Access}, title={A Novel Auto-Focus Method for Image Processing Using Laser Triangulation}, year={2019}, volume={7}, number={}, pages={64837-64843}, doi={10.1109/ACCESS.2019.29141 86}, ISSN={2169-3536}, month={},} 410. 411. 412. [36] A. Hendalianpour et al., "Hybrid Model of IVFRN-BWM and Robust Goal Programming in Agile and Flexible Supply Chain, a Case Study: Automobile Industry," in IEEE Access, vol. 7, pp. 71481-71492, 2019.

doi:

10.1109/ACCESS.2019.2915309 413. 414. @ARTICLE{8713447, author={A. {Hendalianpour} and M. {Fakhrabadi} and X. {Zhang} and M. R. {Feylizadeh} and M. {Gheisari} and P. {Liu} and N. {Ashktorab}}, journal={IEEE Access}, title={Hybrid Model of IVFRN-**BWM and Robust Goal** Programming in Agile and Flexible Supply Chain, a Case Study: Automobile Industry}, year={2019}, volume={7}, number={}, pages={71481-71492}, ISSN={2169-3536}, month={},} 415. 416. 417. [37] M. Gheisari et al., "An **Optimization Model for Software Quality Prediction With Case** Study Analysis Using MATLAB," in IEEE Access, vol. 7, pp. 85123-85138, 2019, doi: 10.1109/ACCESS.2019.2920879. 418. 419. @ARTICLE{8731856,

author={M. {Gheisari} and D. {Panwar} and P. {Tomar} and H. {Harsh} and X. {Zhang} and A. {Solanki} and A. {Nayyar} and J. A. {Alzubi}}, journal={IEEE Access}, title={An Optimization Model for Software Quality Prediction With Case Study Analysis Using

MATLAB},

year={2019}, volume={7}, number={}, pages={85123-85138}, ISSN={2169-3536}, month={}, 420. 421. 422. [38] Noor, F, Sajid, A, Shah, SBH, Zaman, M, Gheisari, M, Mariappan, V. Bayesian estimation and prediction for Burr-Rayleigh mixture model using censored data. Int / Commun Syst. 2019;e4094. https://doi.org/10.1 002/dac.4094 423. 424. @article{doi:10.1002/dac.4 094, 425. author = {Noor, Farzana and Sajid, Ahthasham and Shah, Syed Bilal Hussain and Zaman, Mehwish and Gheisari, Mehdi and Mariappan, Vinayagam}, 426. title = {Bayesian estimation and prediction for Burr-Rayleigh mixture model using censored data}, 427. journal = {International Journal of Communication Systems}, 428. volume = $\{0\}$, 429. number = $\{0\}$, 430. $pages = \{e4094\},\$ 431. } 432. [39] M. Gheisari, Q. Pham, 433. M. Alazab, X. Zhang, C. Fernández-Campusano and G. Srivastava, "ECA: An Edge Computing Architecture for Privacy-Preserving in IoT-Based Smart City," in IEEE Access, vol. 7, pp. 155779-155786, 2019. doi: 10.1109/ACCESS.2019.2937177

434.

435. @ARTICLE{8811469, author={M. {Gheisari} and Q. {Pham} and M. {Alazab} and X. {Zhang} and C. {Fernández-Campusano} and G. {Srivastava}}, journal={IEEE Access}, title={ECA: An Edge Computing Architecture for Privacy-Preserving in IoT-Based Smart City}, year={2019}, volume={7}, number={}, pages={155779-155786}, ISSN={2169-3536}, month={},} 436. [40] M. Gheisari et al., "A 437.

437. [40] M. Gheisan et al., A Survey on Clustering Algorithms in Wireless Sensor Networks: Challenges, Research, and Trends," 2020 International Computer Symposium (ICS), Tainan, Taiwan, 2020, pp. 294-299, doi:

10.1109/ICS51289.2020.00065.

438.

- 439. @INPROCEEDINGS { 9359086 , author={M. {Gheisari} and A. A. {Abbasi} and Z. {Sayari} and Q. {Rizvi} and A. {Asheralieva} and S. {Banu} and F. M. {Awaysheh} and S. B. H. {Shah} and K. A. {Raza}}, booktitle={2020 International Computer Symposium (ICS) }, title={A Survey on Clustering Algorithms in Wireless Sensor Networks: Challenges, Research, and Trends}, year={2020}, volume={}, number={}, pages={294-299}, doi={10.1109/ICS51289.2020.0006 5}} 440.
- 441. **[41]** Kannan, S.; Dhiman, G.; Natarajan, Y.; Sharma, A.; Mohanty, S.N.; Soni, M.; Easwaran, U.; Ghorbani, H.; Asheralieva, A.; Gheisari, M. Ubiquitous Vehicular Ad-Hoc

Network Computing Using Deep Neural Network with IoT-Based Bat Agents for Traffic Management. Electronics 2021, 10, 785. https://doi.org/10.3390/electronics10070785 442. 443. @Article{electronics100707 85, 444. AUTHOR = {Kannan, Srihari and Dhiman, Gaurav and Natarajan, Yuvaraj and Sharma, Ashutosh and Mohanty, Sachi Nandan and Soni, Mukesh and Easwaran, Udayakumar and Ghorbani, Hamidreza and Asheralieva, Alia and Gheisari, Mehdi}, 445. TITLE = {Ubiquitous Vehicular Ad-Hoc Network Computing Using Deep Neural Network with IoT-Based Bat Agents for Traffic Management }, 446. JOURNAL = {Electronics}, 447. $VOLUME = \{10\},\$ 448. $YEAR = \{2021\},\$ 449. NUMBER = $\{7\}$, $ARTICLE-NUMBER = \{785\},$ 450. URL = 451. {https://www.mdpi.com/2079-9292/10/7/785}, 452. } 453. 454. [42] Movassagh, A.A., Alzubi, J.A., Gheisari, M. et al. Artificial neural networks training algorithm integrating invasive weed optimization with differential evolutionary model. J Ambient Intell Human Comput (2021). https://doi.org/10.1007/s12652-020-02623-6 455. 456. @Article{Movassagh2021, 457. = {Ali Akbar author Movassagh and Jafar A. Alzubi and Mehdi Gheisari and Mohamadtaghi Rahimi and Senthilkumar Mohan and Aaqif Afzaal Abbasi and Narjes Nabipour}, 458. journal = {Journal of Ambient Intelligence and Humanized Computing }, 459. title = {Artificial neural networks training algorithm integrating invasive weed optimization with differential evolutionary model}, 460. year $= \{2021\},\$ 461. month $= \{ mar \},\$

462. doi = {10.1007/s12652-020-02623-6}, 463. publisher = {Springer Science and Business Media $\{LLC\}\},\$ 464. } 465. 466. [43] Shao, Yongfu, et al. "Optimization of Ultrasound Information Imaging Algorithm in Cardiovascular Disease Based on Image Enhancement." Mathematical Problems in Engineering 2021 (2021). 467. 468. @article{shao2021optimizat ion, 469. title={Optimization of Ultrasound Information Imaging Algorithm in Cardiovascular Disease Based on Image Enhancement } , author={Shao, Yongfu and 470. Wu, Jue and Ou, Hongping and Pei, Min and Liu, Li and Movassagh, Ali Akbar and Sharma, Ashutosh and Dhiman, Gaurav and Gheisari, Mehdi and Asheralieva, Alia}, 471. journal={Mathematical Problems in Engineering}, volume={2021}, 472. 473. year={2021}, 474. publisher={Hindawi} 475. } 476. 477. [44] Gheisari, M., Najafabadi, H. E., Alzubi, J. A., Gao, J., Wang, G., Abbasi, A. A., & Castiglione, A. (2021). OBPP: An ontology-based framework for privacy-preserving in IoT-based smart city. Future Generation Computer Systems. 478. 479. @article{GHEISARI20211, 480. title = {OBPP: An ontology-based framework for privacy-preserving in IoT-based smart city}, 481. journal = {Future Generation Computer Systems}, 482. $volume = \{123\},\$ 483. pages = $\{1-13\}$, 484. year = $\{2021\},\$ 485. $issn = \{0167 - 739X\},\$ 486. author = {Mehdi Gheisari and Hamid Esmaeili Najafabadi

and Jafar A. Alzubi and Jiechao Gao and Guojun Wang and Aaqif Afzaal Abbasi and Aniello Castiglione}, 487. } 488. 489. [45] Hosseini Bamakan S M, Rahbar E, Gheisari M. Role of Wearable Technology in the Diagnosis and Prevention of COVID-19. J Research Health. 2021; 11 (4) :1-1 490. 491. @ARTICLE{Hosseini Bamakan, 492. author = {Hosseini Bamakan, Seyed Mojtaba and Rahbar, Ehsan and Gheisari, Mehdi and }, 493. title = {Role of Wearable Technology in the Diagnosis and Prevention of COVID-19}, 494. $volume = \{11\},\$ 495. number = $\{4\}$, 496. eprint = {http://jrh.gmu.ac.ir/article-1-1961-en.docx}, journal = {Journal of 497. Research and Health}, 498. doi = {10.32598/JRH.11.4.1903.1}, 499. $year = \{2021\}$ 500. } 501. 502. [46] S. Afrasiabi, B. Behdani, M. Afrasiabi, M. Mohammadi, A. Asheralieva and M. Gheisari. "Differential Protection of Power Transformers based on RSLVQ-Gradient Approach Considering SFCL," 2021 IEEE Madrid PowerTech, 2021, pp. 1-6, doi: 10.1109/PowerTech46648.2021.9494873. 503. 504. @INPROCEEDINGS { 9494873 , author={Afrasiabi, Shahabodin and Behdani, Behzad and Afrasiabi, Mousa and Mohammadi, Mohammad and Asheralieva, Alia and Gheisari, Mehdi}, booktitle={2021 IEEE Madrid PowerTech} , title={Differential Protection of Power Transformers based on RSLVQ-Gradient Approach Considering SFCL}, year={2021}, volume={}, number={}, pages={1-6}, doi={10.1109/PowerTech46648.202 1.9494873} 505.

506. [47] B. Behdani, M. Allahbakhshi, A. Asheralieva and M. Gheisari. "Analytical Method for Ferroresonance Solutions in Series Compensated Power Systems due to GICs: A Graphical Approach," 2021 IEEE Madrid *PowerTech*, 2021, pp. 1-6, doi: 10.1109/PowerTech46648.2021.9494921. 507 508. @INPROCEEDINGS { 9494921, author={Behdani, Behzad and Allahbakhshi, Mehdi and Asheralieva, Alia and Gheisari, Mehdi}, booktitle={2021 IEEE Madrid PowerTech}, title={Analytical Method for Ferroresonance Solutions in Series Compensated Power Systems due to GICs: A Graphical Approach}, year={2021}, volume={}, number={}, pages= $\{1-6\}$, doi={10.1109/PowerTech46648.202 $1.9494921\}$ 509. 510. [48] Natarajan, Y., et al.: An IoT and machine learning-based routing protocol for reconfigurable engineering application. IET *Commun.* 00, 1– 12 (2021). https://doi. org/10.1049/cmu2.12266 511. 512. 513. @Article{Natarajan2021, 514. author = {Yuvaraj Natarajan and Kannan Srihari and Gaurav Dhiman and Selvaraj Chandragandhi and Mehdi Gheisari and Yang Liu and Cheng-Chi Lee and Krishna Kant Singh and Kusum Yadav and Hadeel Fahad Alharbi}, 515. journal $= \{ \{ IET \} \}$ Communications}, 516. title $= \{An \{IoT\}\}$ and machine learning-based routing protocol for reconfigurable engineering application}, 517. year $= \{2021\},\$ 518. month $= \{aug\},\$ 519. doi $\{10.1049/cmu2.12266\},\$ 520. publisher = {Institution of Engineering and Technology ({IET})}, 521. } 522.

523. [49] Mangla, Monika, et al. "A **Proposed Framework for Autonomic Resource Management in Cloud** Computing Environment." Autonomic Computing in Cloud Resource Management in Industry 4.0. Springer, Cham, 2021. 177-193. 524 525. @incollection{mangla2021pr oposed, title={A Proposed 526. Framework for Autonomic Resource Management in Cloud Computing Environment }, 527. author={Mangla, Monika and Deokar, Sanjivani and Akhare, Rakhi and Gheisari, Mehdi}, 528. booktitle={Autonomic Computing in Cloud Resource Management in Industry 4.0}, 529. $pages = \{177 - -193\},\$ 530. year={2021}, 531. publisher={Springer} 532. } 533. [50] Li, Lintao, et al. "Research on 534. TCP Performance Model and Transport Agent Architecture in Broadband Wireless Network." Scalable Computing: Practice and Experience 22.2 (2021): 193-201. 535. @article{li2021research, 536. title={Research on TCP 537. Performance Model and Transport Agent Architecture in Broadband Wireless Network}, 538. author={Li, Lintao and Sharma, Parv and Gheisari, Mehdi and Sharma, Amit}, journal={Scalable 539. Computing: Practice and Experience}, 540. volume={22}, 541. number= $\{2\}$, $pages = \{193 - -201\},\$ 542. 543. year={2021} 544. 545. [51] S. Afrasiabi, B. Behdani, M. Afrasiabi, M. Mohammadi, Y. Liu and M. Gheisari, "A Comparative Analysis of Artificial Intelligence for Power Transformer Differential Protection." 2021 IEEE International Conference on Environment and Electrical Engineering and 2021 IEEE Industrial

and Commercial Power Systems Europe (EEEIC / I&CPS Europe), 2021, pp. 1-6, doi: 10.1109/EEEIC/ICPSEurope51590.2021. 9611033.

546. 547.

@INPROCEEDINGS{9611033,

- 548. author={Afrasiabi, Shahabodin and Behdani, Behzad and Afrasiabi, Mousa and Mohammadi, Mohammad and Liu, Yang and Gheisari, Mehdi},
- 549. booktitle={2021 IEEE International Conference on Environment and Electrical Engineering and 2021 IEEE Industrial and Commercial Power Systems Europe (EEEIC / I CPS Europe)},
- 550. title={A Comparative Analysis of Artificial Intelligence for Power Transformer Differential Protection},
- 551. year={2021},
- 552. volume={},
- 553. number={}.
- 554. pages={1-6},

555.

doi={10.1109/EEEIC/ICPSEurope51590.2 021.9611033}}

556.

557. [52] M. Pazhoohesh, M. S. Javadi,
M. Gheisari, S. Aziz and R. Villa, "Dealing with Missing Data in the Smart Buildings using Innovative Imputation Techniques," *IECON 2021 – 47th Annual Conference of the IEEE Industrial Electronics Society*, 2021, pp. 1-7, doi:

10.1109/IECON48115.2021.9612650.

558.

559.

- 560. @INPROCEEDINGS{9612650,
- 561. author={Pazhoohesh, Mehdi and Javadi, Mohammad Sadegh and Gheisari, Mehdi and Aziz, Saddam and Villa, Raffaella},
- 562. booktitle={IECON 2021 47th Annual Conference of the IEEE Industrial Electronics Society}.
- 563. title={Dealing with Missing Data in the Smart Buildings using Innovative Imputation Techniques},

564. year={2021},

- 565. volume={},
- 566. number={},
- 567. pages={1-7},

568.

doi={10.1109/IECON48115.2021.9612650 }}

569.

570. [53] K. A. Raza, A. Asheralieva, M. M. Karim, K. Sharif, M. Gheisari and S. Khan, "A Novel Forwarding and Caching Scheme for Information-Centric Software-Defined Networks," 2021 International Symposium on Networks, Computers and Communications (ISNCC), 2021, pp. 1-8, doi:

10.1109/ISNCC52172.2021.9615667.

- 571.
- 572. @INPROCEEDINGS{9615667, author={Raza, Khuhawar Arif and Asheralieva, Alia and Karim, Md Monjurul and Sharif, Kashif and Gheisari, Mehdi and Khan, Salabat}, booktitle={2021 International Symposium on Networks, Computers and Communications (ISNCC)}, title={A Novel Forwarding and Caching Scheme for Information-Centric Software-Defined Networks}, year={2021}, volume={}, number={}, pages={1-8}, doi={10.1109/ISNCC52172.2021.9615667

doi={10.1109/ISNCC52172.2021.9615667
}}

- 573.
- 574. [54] Yogesh Kumar, Apeksha Koul, Pushpendra Singh Sisodia, Jana Shafi, Verma Kavita, Mehdi Gheisari, Mohamad Bagher Davoodi, "Heart Failure Detection Using Quantum-Enhanced Machine Learning and Traditional Machine Learning Techniques for Internet of Artificially Intelligent Medical Things", Wireless Communications and Mobile Computing, vol. 2021, Article ID 1616725, 16 pages, 2021.
- 575.
- 576. @Article{Kumar2021,
- 577. author = {Yogesh Kumar and Apeksha Koul and Pushpendra Singh Sisodia and Jana Shafi and Verma Kavita and Mehdi Gheisari and Mohamad Bagher Davoodi},
- 578. journal = {Wireless Communications and Mobile Computing},
- 579. title = {Heart Failure Detection Using Quantum-Enhanced Machine Learning and Traditional Machine

Learning Techniques for Internet of Artificially Intelligent Medical Things}, 580. $= \{2021\},\$ vear 581. month $= \{ dec \},\$ 582. pages = $\{1-16\}$, 583. volume = $\{2021\}$, 584. $= \{10.1155/2021/1616725\},\$ doi editor = {Mohammad R 585. Khosravi}, 586. publisher = {Hindawi Limited}, 587. } 588. 589. [55] 590.@book{gupta2021cancer, 591. title={Chapter 7th, Cancer Prediction for Industrial IoT 4.0: A Machine Learning Perspective}, 592. author={Ashish Kumar, Revant Singh Rai, Mehdi Gheisari}, 593. isbn={9781000508581}, series={Prediction of 594. cervical cancer using machine learning}, 595. url={https://books.google.co. in/books?id=vDxOEAAAQBAJ}, 596. year={2021}, publisher={CRC Press} 597. 598.} 599. 600. [56] Gheisari, Mehdi, et al. "A novel enhanced algorithm for efficient human tracking." Int J Inf & Commun Technol 11.1 (2022): 1-7. @article{gheisari2022novel 601. 602. title={A novel enhanced algorithm for efficient human tracking}, author={Gheisari, Mehdi 603. and Safari, Zohreh and Almasi, Mohammad and Sridharan, Abel and GK, Ragesh and Liu, Yang and Abbasi, Aaqif Afzaal}, journal={Int J Inf \& 604. Commun Technol}, 605. volume={11}, 606. number= $\{1\}$, pages= $\{1--7\}$, 607. 608. year={2022} 609. } 610. 611. [57] Abdullah Ajmal, Hamza

Bader Alouffi, Mehdi Gheisari, "Stress-Relieving Video Game and Its Effects: A POMS Case Study", Computational Intelligence and Neuroscience, vol. 2022, Article ID 4239536, 11 pages, 2022. https://doi.org/10.1155/2022/4239536 612. 613. [58] Yongsheng Rao, Saeed Kosari, Mehdi Gheisari, "New Results in Vague **Incidence Graphs with** Application", Journal of Function Spaces, vol. 2022, Article ID 3475536, 7 pages, 2022. https://doi.or g/10.1155/2022/3475536 614. 615. @article{rao2022new, 616. title={New Results in Vague Incidence Graphs with Application }, 617. author={Rao, Yongsheng and Kosari, Saeed and Gheisari, Mehdi}, journal={Journal of 618. Function Spaces }, 619. volume={2022}, 620. $year = \{2022\},\$ 621. publisher={Hindawi} 622. } 623. 624. 625. [59] J. A. Alzubi *et al.*, "A Dynamic SDN-based Privacy-Preserving Approach for Smart City Using Trust Technique," 2022 9th Iranian Joint Congress on Fuzzy and Intelligent Systems (CFIS), 2022, pp. 1-5, doi: 10.1109/CFIS54774.2022.9756458. 626. @INPROCEEDINGS{9756458. author={Alzubi, Jafar A. and 627.

626. @INPROCEEDINGS{9756458,
627. author={Alzubi, Jafar A. and Movassagh, AliAkbar and Gheisari, Mehdi and Najafabadi, Hamid Esmaeili and Abbasi, Aaqif Afzaal and Liu, Yang and Pingmei, Zhou and Izadpanahkakhk, Mahdieh and Najafabadi, AmirHossein Pourishaban},
628. booktitle={2022 9th Iranian Joint Congress on Fuzzy and Intelligent Systems (CFIS)},
629. title={A Dynamic SDN-based

Privacy-Preserving Approach for Smart City Using Trust Technique},

- 630. year={2022},
- 631. volume={},
- 632. number={},
- 633. pages={1-5},

Aldabbas, Rashid Amin, Sundas Ibrar,

634.

doi={10.1109/CFIS54774.2022.9756458}}

635. [60] Yongsheng Rao, Saeed Kosari, Mehdi Gheisari, "New Results in Vague Incidence Graphs with Application", *Journal of Function Spaces*, vol. 2022, Article ID 3475536, 7 pages, 2022. https://doi .org/10.1155/2022/3475536

636.

643.

}

- 637. [61] Moshayedi, Ata Jahangir et al. "Automation Attendance Systems Approaches: A Practical Review." BOHR International Journal of Internet of Things Research (2021): n. pag.
- 638.@article{Moshayedi2021Automa tionAS,
- 639. title={Automation Attendance Systems Approaches: A Practical Review},
- 640. author={Ata Jahangir Moshayedi and Atanu Roy and Liefa Liao and Mehdi Gheisari and Aaqif Afzaal Abbasi and Seyed Mojtaba Hosseini Bamakan},

```
641. journal={BOHR
    International Journal of
    Internet of Things Research},
642. year={2021}
```

